Author’s response to reviews

Title: A high-throughput protocol for mutation scanning of the BRCA1 and BRCA2 genes.

Authors:

Heather L Hondow (heather.hondow@petermac.org)
Stephen B Fox (stephen.fox@petermac.org)
Gillian Mitchell (gillian.mitchell@petermac.org)
Rodney J Scott (Rodney.Scott@newcastle.edu.au)
Victoria Beshay (victoria.beshay@petermac.org)
Stephen Q Wong (stephen.wong@petermac.org)
kConFab investigators (heather.thorne@petermac.org)
Alexander Dobrovic (alexander.dobrovic@petermac.org)

Version: 4 Date: 17 June 2011

Author’s response to reviews:

The changes indicated by the journal editorial staff have been made.

The title has been abbreviated.

Typographical errors have been corrected and small final corrections and revisions, which have helped the readability of the MS but have not otherwise affected the content, have been made.